Author's response to reviews

Title: An association between Helicobacter pylori infection and cognitive function in children at early school age: a community-based study

Authors:

Khitam Muhsen (muhsenkh@post.tau.ac.il)
Asher Ornoy (ornoy@huji.ac.il)
Ashraf Akawi (ashaka74@walla.com)
Gershon Alpert (alpert@clalit.org.il)
Dani Cohen (dancohen@post.tau.ac.il)

Version: 6 Date: 28 April 2011

Author's response to reviews: see over
April 28, 2011

To: The BMC Series Editorial Production Team

Subject: Editing and formatting changes MS: 1191253392448967.

Enclosed please find a revised version of our manuscript (BMC Pediatrics MS: 1191253392448967).

We have edited the manuscript according to your requirements. The details of these changes are:

1. Abstract: This should be identical in the manuscript and the submission system.
   
   **Reply:** We revised the abstract in the manuscript to identical to that in the submission system.

2. Authors' contributions: Please indicate that all authors read and approved the final manuscript.
   
   **Reply:** We added this sentence into the "authors' contributions" section.

3. References: please provide a functioning link to website reference number 25
   
   **Reply:** We provided a functioning link to the website of reference number 25.

4. Email address: please include email addresses for all the authors on the title page, using the following format:
   AB: abcd@institution.ac.uk
   EF: efgh@generic.co.uk
   IJ: IJKL@corporation.com
   
   **Reply:** We edited the email addresses on the title page following the required format.

5. Line numbers: Please remove these
   
   **Reply:** Line numbers were removed.

6. Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.
   
   **Reply:** Highlighting and tracking were removed.
7. Tables: please remove the visible vertical lines from your tables. Please note that we are unable to correctly display merged cells where the merged cell crosses rows: please re-layout your table without these merged elements.

Reply: The visible vertical lines were removed from the tables. We also unmerged the cells in the tables.

8. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Reply: We checked typography errors in the manuscripts and corrected these errors. Thank you for this opportunity.

Thank you.

Khitam Muhsen

Tel Aviv University